Author's response to reviews

Title: Job satisfaction trends during nurses' early career

Authors:

    Trevor Murrells (trevor.murrells@kcl.ac.uk)
    Sarah Robinson (sarah.robinson@kcl.ac.uk)
    Peter Griffiths (peter.griffiths@kcl.ac.uk)

Version: 5 Date: 14 May 2008

Author's response to reviews:

To: Editor-in-Chief

Dear Dr Norton

Table 2 has been reformatted to remove gridlines previously shown in grey.

Regards

Trevor Murrells
National Nursing Research Unit
King's College London